Author's response to reviews

Title: Design and methods for a randomized clinical trial of a diabetes self-management intervention for low-income Latinos: Latinos en Control

Authors:

    Milagros C Rosal (Milagros.Rosal@umassmed.edu)
    Mary Jo White (Maryjo.White@umassmed.edu)
    Angela Restrepo (Angela.Restrepp@umassmemorial.org)
    Barbara Olendzki (Barbara.Olendzki@umassmed.edu)
    Jeffrey Scavron (Jeffrey.Scavron@bhs.org)
    Elise Sinagra (EliseSinagra@msn.com)
    Ira S Ockene (Ira.Ockene@umassmed.edu)
    Michael Thompson (mthompson@mfa.gwu.edu)
    Stephenie C Lemon (Stephenie.Lemon@umassmed.edu)
    Lucy M Candib (Lucy.CandibFHCW@umassmed.edu)
    George Reed (George.Reed@umassmed.edu)

Version: 4 Date: 30 November 2009

Author's response to reviews:

Dear Editor:

We have made 2 minor revisions to the accepted manuscript.

We have added a middle initial to the name of one of the authors.

Thus the name of this author in the title page now reads Lucy M. Candib.

In the discussion section, 4th line it read low literacy needs, language barriers....

We have removed the word [needs].

Thank you,

Milagros Rosal